Dear reviewer and editor,

We would like to express our gratitude to the editors and reviewers for your precious time and advice. Accordingly we have revised the manuscript. We hope that these revisions and adjustments are satisfactory and that the revised manuscript meets the publishing requirements of the World Journal of Gastrointestinal Surgery. Any question will be highly welcomed.

Reviewer #1:
Scientific Quality: Grade B (Very good)
Language Quality: Grade A (Priority publishing)
Conclusion: Minor revision
Specific Comments to Authors:
1 Title. ok
2 Abstract. the abstract summarize and reflect the work described in the manuscript but conclusion is not adapted
Response: We have revised the conclusion in the abstract. More details were in the manuscript.
3 Key words. Ok
4 case presentation very interesting but very long. please synthesize more precision about surgery and post operative outcome wath about genetic exam?
Response: We have simplified case presentation in the manuscript. During the entire treatment, we recommended the patient further genetic exam while she refused which is a minor flaw of this research.
7 Discussion. informative and correctly written
8 Illustrations and tables. Very good illustration
9 11 References ok
Response: Thank you for your approval.

Science editor:
In this case report, a patient with CCS complicated by intussusception is presented. The authors should revise the manuscript according to the comments of the referee. Especially the discussion section should be comprehensively revised. Please avoid state under headings in the discussion section. Similarly, avoid the repetitions throughout the text. Provide Copyright License Agreement. Please upload the PDF version of the Copyright License Agreement Form and send it after all authors have signed it. Conflict-of-Interest Disclosure Form should be provided.
Language Quality: Grade B (Minor language polishing)
Scientific Quality: Grade B (Very good)
Response: Thanks for your suggestion. We have revised the manuscript as required.

Best regards
Sincerely yours,
You-Wei Chen
Zhejiang Provincial People's Hospital, People's Hospital of Hangzhou Medical College
E-mail: cyw@zju.edu.cn